

Reassessing the distribution of Burkholderia pseudomallei outside known endemic areas using animal serological screening combined with environmental surveys: The case of Les Saintes (Guadeloupe) and French Guiana

Mégane Gasqué, Vanina Guernier-Cambert, Gil Manuel, Rachid Aaziz, Jules Terret, Thomas Deshayes, Xavier Baudrimont, Sebastien Breurec, Emma Rochelle-Newall, Karine Laroucau

▶ To cite this version:

Mégane Gasqué, Vanina Guernier-Cambert, Gil Manuel, Rachid Aaziz, Jules Terret, et al.. Reassessing the distribution of Burkholderia pseudomallei outside known endemic areas using animal serological screening combined with environmental surveys: The case of Les Saintes (Guadeloupe) and French Guiana. PLoS Neglected Tropical Diseases, 2024, 18 (9), pp.e0011977. 10.1371/journal.pntd.0011977. ird-04791493

HAL Id: ird-04791493 https://ird.hal.science/ird-04791493v1

Submitted on 19 Nov 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



GOPEN ACCESS

Citation: Gasqué M, Guernier-Cambert V, Manuel G, Aaziz R, Terret J, Deshayes T, et al. (2024) Reassessing the distribution of *Burkholderia pseudomallei* outside known endemic areas using animal serological screening combined with environmental surveys: The case of Les Saintes (Guadeloupe) and French Guiana. PLoS Negl Trop Dis 18(9): e0011977. https://doi.org/10.1371/ journal.pntd.0011977

Editor: Elizabeth J. Carlton, University of Colorado -Anschutz Medical Campus, UNITED STATES OF AMERICA

Received: February 7, 2024

Accepted: August 28, 2024

Published: September 26, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0011977

Copyright: © 2024 Gasqué et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

RESEARCH ARTICLE

Reassessing the distribution of *Burkholderia pseudomallei* outside known endemic areas using animal serological screening combined with environmental surveys: The case of Les Saintes (Guadeloupe) and French Guiana

Mégane Gasqué^{1,2}, Vanina Guernier-Cambert^{2,3}*, Gil Manuel⁴, Rachid Aaziz¹, Jules Terret¹, Thomas Deshayes¹, Xavier Baudrimont⁵, Sébastien Breurec⁶, Emma Rochelle-Newall⁶, Karine Laroucau¹

1 Animal health laboratory, Bacterial Zoonosis Laboratory, Anses, Paris, France, 2 Sorbonne Université, UPEC, IRD, CNRS, INRAE, Institute of Ecology and Environmental Sciences-Paris (iEES-Paris), Paris, France, 3 Faculty of Veterinary Technology, Kasetsart University, Bangkok, Thailand, 4 Veterinary clinic of Carmel, Guadeloupe, France, 5 Unit for Animal and Plant Health and Protection Environment (SPAVE), Directorate for Environment, Agriculture, Food and Forestry (DEAAF), Veterinary and Phytosanitary Inspection Service (SIVEP), French Guiana, France, 6 Pasteur Institute of Guadeloupe, Morne Jolivière, Guadeloupe, France

* vanina.guernier@ird.fr

Abstract

Background

Melioidosis, an emerging infectious disease that affects both humans and animals, is caused by the soil-dwelling bacterium *Burkholderia pseudomallei*. It is endemic in South and Southeast Asia, and northern Australia, causing an estimated 165,000 human cases annually worldwide. Human cases have been reported in the French West Indies (Martinique and Guadeloupe) since the 1990s. Conversely, no human cases have been reported in French Guiana, a French territory in South America. Our study aimed to investigate whether *B. pseudomallei* is locally established in Guadeloupe and French Guiana using animals as a proxy.

Methodology/principal findings

Blood samples were collected from different animals from 56 farms in French Guiana (n = 670) and from two goat farms in Les Saintes (n = 31), *part of the Guadeloupe archipelago* and tested by enzyme-linked immunosorbent assay (ELISA). In Les Saintes, a serological follow-up was performed, and soil, water and goat rectal swabs were collected and analyzed by culture and PCR. The highest seroprevalence rates (39%) were observed in goats in Les Saintes, followed by horses (24%) and cattle (16%) in French Guiana. In the two goat farms, supplementary analyses detected *B. pseudomallei* from one goat rectal swab, and a *B. pseudomallei* strain was isolated from the soil.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This study was supported by the lle de France Region (Dim1Health) (ERN, MG) and the European Commission Directorate-General for Health and Consumers (EC no.180/2008) (KL). We thank the French Institut de Recherche pour le Développement (IRD) and the French Agence nationale de sécurité sanitaire de l'alimentation, de l'environnement et du travail (ANSES) for their financial support. The funders had no role in the design of the study, collection and analysis of the data, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Conclusions/significance

Our animal serological data suggest the presence of *B. pseudomallei* in Les Saintes and French Guiana. In Les Saintes, environmental surveys confirmed the endemicity of the bacteria, which is consistent with documented human cases of melioidosis on the island. We did not conduct an environmental survey in French Guiana. Nevertheless, our serological results call for local environmental surveys and a retrospective reassessment of human infections with melioidosis-like symptoms.

Author summary

Melioidosis, a disease caused by the environmental bacterium *Burkholderia pseudomallei*, has historically been described in Southeast Asia and northern Australia. However, recent studies have demonstrated its presence outside these areas, both in the environment and in patients without a history of travel to known endemic areas. In addition, the predicted increase in extreme climatic events in the near future could increase the prevalence of the disease and lead to its emergence in new areas. For these reasons, it is important to identify areas at risk outside of known endemic areas.

Hypothesizing that we would have little chance of finding *B. pseudomallei* through random environmental surveys, we used serological testing to find evidence of past exposure to the bacteria in apparently healthy domestic animals. We identified seropositive animals in Les Saintes and French Guiana. We then searched for the presence of *B. pseudomallei* in the immediate vicinity of the seropositive animals in Les Saintes, and isolated it in the soil of a goat farm. Our study suggests that domestic animals could be used as sentinels for the detection of melioidosis outside of countries with frequent human cases.

Introduction

Melioidosis is an opportunistic infectious disease with a significant prevalence in South Asia, Southeast Asia, and northern Australia, primarily due to environmental exposure to *Burkholderia pseudomallei*. This soil-dwelling bacterium can affect both humans and animals through inhalation, ingestion, or skin contact [1]. *B. pseudomallei* is listed as a Class 1 pathogen by the US Centers for Disease Control and Prevention (CDC) and as a selected agent under the Highly Pathogenic Microorganisms and Toxins (MOT) regulation by the French National Agency for the Safety of Medicines and Health Products (ANSM).

The global burden of melioidosis was predicted to be 165,000 human cases in 2015 [2]. In addition, reports of human and, occasionally, animal cases are increasing along with environmental evidence of the bacteria worldwide, highlighting the likely presence of the bacterium on all continents, including Africa [3,4], North America [5,6], and South and Central America [7–9].

The clinical presentation of the disease is highly variable between individuals and species, ranging from chronic forms to acute septicemia [1]. Diagnosis and treatment of human melioidosis are complicated by the lack of pathognomonic clinical features and the antibiotic resistance of the bacterium to several antibiotics. Risk factors such as diabetes and chronic diseases contribute to increased susceptibility [10]. In animals, melioidosis affects a variety of species with multiple cases documented in domestic animals such as cats, cattle, deer, dogs, goats,

equids, sheep, and sporadic cases in other species, including marine mammals, birds and even crocodiles in zoos or wildlife parks [11].

While accurate epidemiological data and diagnostic tools are lacking for human melioidosis, they are even more limited for animal melioidosis. Diagnostic approaches for animal melioidosis are generally the same as for humans [11–13]. Although diagnostic methods such as indirect hemagglutination (IHA), complement fixation test (CFT) and enzyme-linked immunosorbent assay (ELISA) have been developed for commonly tested species such as small ruminants, pigs, and equids [14–16], standardized tests for the diagnosis of melioidosis in all susceptible species are currently lacking. A recent study demonstrated the effectiveness of serological tests developed for glanders (a disease caused by *B. mallei*, a bacterium closely related to *B. pseudomallei*) in accurately detecting melioidosis in equids [17]. These tests include both a reference method (CFT) and alternative methods (ELISA, Western blot).

The bacterium *B. pseudomallei* is mainly found in moist clay soils and in turbid water [18,19]. Its distribution in soils is strongly influenced by climatic events, with more cases observed during the rainy season and sporadic contamination peaking after extreme weather events such as storms or floods [20,21]. Environmental detection of *B. pseudomallei* is challenging and specific culture protocols are required to isolate the bacterium [22]. Detection may be more difficult in areas of low endemicity or emergence, where the bacterium may be present at lower concentrations and/or in limited locations. A new culture medium using erythritol as a carbon source has recently been developed to improve culture isolation from soil samples [23]. It was successfully used to isolate *B. pseudomallei* from a rice farm in south central Ghana, West Africa [24].

Cases of melioidosis have been reported in both South, Central, and North America. This includes the French West Indies, i.e., Guadeloupe and Martinique, where 21 human cases have been reported since 1993 [25–30], some of whom had no history of travel outside of their island of residence. In contrast, no human cases have been reported in French Guiana (Guiana Shield, South America), while human cases have been reported in neighboring Northeastern Brazil [9].

The first objective of this study was to assess the exposure of different livestock species (goats, cattle, horses, sheep) to *B. pseudomallei* in Les Saintes, a group of islands in the Guade-loupe archipelago, and in French Guiana. The second objective was to confirm the local presence of the bacteria in farms with seropositive animals by molecular testing of fecal and environmental samples, and isolation of *B. pseudomallei* by culture.

Materials and methods

Ethics statement

Animal sampling protocols have been approved by the Institute of Research and Development (IRD) Ethics Committee (April 21, 2022). The application submitted to the Ethics Committee covers blood and swab collection by veterinarians on domestic animals in Les Saintes and French Guiana for the period January 1, 2021 to December 31, 2023. Written consent was also obtained from the two goat breeders in Les Saintes. In addition, a Nagoya Protocol dossier (European level, National Decree No. 2017–848 of May 9, 2017) was approved for animal and environmental samplings. Anses is authorized to handle *B. pseudomallei* in accordance with the ANSM regulation on MOT, and to import, analyze and store environmental samples in accordance with the EU Regulation 2019–829 (DAAF, Prefectural Order of 25 August 2022).

Initial animal sampling in French Guiana and Les Saintes

A total of 56 farms were sampled in French Guiana, mainly along the north coast where 80% of the population lives (Fig 1B). These farms included both single-species and multi-species



Fig 1. Map of sampling sites. Blue dots represent farms where animals were tested in (A) Les Saintes (n = 2) and (B) French Guiana (n = 56). Les Saintes is part of the Guadeloupe archipelago, and consists of nine islands, of which only the two largest are inhabited, Terre-De-Haut and Terre-De-Bas; only Terre-De-Haut was sampled. The map layers (country boundaries) of South America, Martinique and French Guiana were downloaded from the geoBoundaries website https://www.geoboundaries.org/simplifiedDownloads.html).

farms, ranging from 2 to 31 animals per farm. Among the different regions of French Guiana, *18 farms were located in Macouria, 11 in Kourou, 7 in Mana, 7 in Saint Laurent du Maroni, 4 in Sinnamary, 3 in Montsinery Tonnegrande, 2 in Cayenne, and 1 in Iracoubo, Matoury Roura, and Saint George de l'Oyapock*. Notably, 10 of these farms had more than one species, including cattle, sheep and horses. Blood samples were collected between March 2021 and September 2023 as part of an annual surveillance program conducted by the Direction of Food, agriculture and forests (DAAF—Direction de l'Alimentation, de l'Agriculture et de la Forêt) in French Guiana. A subsample of this serobank was provided by the DAAF veterinarian and included (n = 670): 361 cattle from 37 farms, 131 sheep from 15 farms, 100 goats from 10 farms, 63 equines from six farms, and 15 pigs from one farm.

In Les Saintes, on the island of Terre-De-Haut, two goat farms were included in this study and monitored by a local veterinarian: Farm A housed approximately 40 goats and Farm B housed approximately 60 goats. Sampling was not exhaustive as they were free roaming (Fig 1A). On both farms, mixed-sex goats grazed freely in open pastures, consuming local vegetation, and no birth control measures were used, hence the approximate herd sizes reported by the farmers. The topography of the pastures was characterized by steep terrain and both were susceptible to significant surface water runoff. In November 2021, 31 goats were sampled for blood: 14 from Farm A and 17 from Farm B. After blood clotting and centrifugation at 900 rpm for 10 min, sera were collected, heat-inactivated at 60°C for 30 min and stored at -20°C until analysis. These two farms were selected for further investigation based on their accessibility and on previous reports of human cases in Les Saintes.

Serological analysis

GLANDA ELISA. The ID ScreenGlanders Double Antigen Multi-species ELISA test (Innovative Diagnostics, Grabels, France) (hereafter referred to as GLANDA ELISA) was used according to the manufacturer's instructions. This test does not rely on the use of a specific conjugate and can therefore be used for multiple animal species. The relative amounts of antibodies in the serum samples were calculated with reference to the positive and the negative

controls provided in the test. The serological titer (expressed as the sample to positive ratio or S/P%) was calculated from the optical density (OD) reading at 450 nm using the modified formula: S/P% = ((OD sample—OD negative control) / (OD positive control—OD negative control)) x 100. Serum samples with S/P% <70% were considered negative and \geq 70% were considered positive.

The GLANDA ELISA test was originally developed for the diagnosis of glanders. A *B. pseudomallei* seropositivity result using this commercial test is therefore considered as presumptive rather than absolute; all seropositivity rates in our study were calculated based on these presumptive positive numbers.

Additional serological tests on equine sera, French Guiana. The complement fixation test (CFT), the reference test for equine glanders, was performed as previously described using the cold method with the Malleus CFT antigen and the associated positive and negative controls (Bioveta, Czech Republic) [31]. Serum samples were initially tested at 1/5 dilution. Samples with 100% hemolysis were considered negative, those with 25–75% hemolysis were considered equivocal, and those with 100% inhibition of hemolysis were considered positive. All samples initially identified as equivocal or positive were retested over a range of dilutions. The "titer" is a six-digit barcode corresponding to the intensity of hemolysis inhibition (0 = 0%; 1 = 25%; 2 = 50%; 3 = 75%; 4 = 100%) at the reciprocal dilutions (1/5, 1/10, 1/20, 1/40, 1/80, 1/160) for each sample. Anti-complement activity (due to incomplete elimination of complement proteins during the serum heat inactivation) was checked for each serum.

In addition, the Luminex assay, originally developed for the diagnosis of equine glanders, was performed as previously described using the heat shock protein (GroEL) (BPSL2697) and hemolysin-coregulated protein (Hcp1) (BPSS1498) antigens [32]. A serum from a horse naturally infected with *B. mallei* (identified as MRI#1) was used as a positive control to determine S/P%, which was calculated for each antigen using the same formula as for GLANDA ELISA. S/P% values greater than 45% for GroEL and 43% for Hcp1 were considered positive, as previously established [31].

Follow-up sampling of selected GLANDA ELISA positive farms. Additional serological monitoring was carried out on selected farms with GLANDA ELISA positive animals. In French Guiana, additional serological testing was carried out on a horse farm in January 2022 and a more comprehensive longitudinal study was conducted on two goat farms in Les Saintes.

Regarding the longitudinal study in Les Saintes, after the initial sampling in November 2021 (described above), blood samples and rectal swabs or feces were collected by the local veterinarian from April 2022 to December 2022 on farm B, and from April 2022 to March 2023 on farm A (Fig 2). Due to the free-range nature of the farms, goat capture was not exhaustive during each field visit. In April 2022, as many animals as possible were captured, sampled and identified with a plastic collar. However, due to the loss of these plastic collars, microchip reidentification was conducted in August 2022. This corresponds to sampling #1 of the longitudinal study (sampling #2 occurred in December 2022, sampling #3 in March 2023; see Fig 2). Blood samples were processed identically to the initial visit, and rectal swabs were stored dry at -20°C until PCR analysis.

Environmental sampling at the two goat farms in Les Saintes

Soil and water samples were collected from both goat pastures (farms A and B) in December 2022, i.e., during the rainy season (Fig 2).

Soil samples

Soil samples (n = 50 in farm A, n = 42 in farm B) were collected along transects perpendicular to the main slope of the pasture. Transect lines were at least 6 m apart and sampling points





along the same transect line were 5 m apart. Samples were taken at a depth of 30 cm using a standard soil auger, which was disinfected with absolute ethanol after each sampling. Approximately 200 g of soil was collected and placed in sealed Ziploc bags to preserve moisture. Samples were stored in the dark and at room temperature (approximately 25°C) until shipment to France. Samples were analyzed within two weeks of collection. Each soil sample was homogenized manually and 10 g of soil was subsampled and added to 10 mL of sterile water. After vigorous vortexing, the suspension was shaken on an orbital shaker at 160 rpm for 1 h and then allowed to settle for 10 min. A 1 mL sample of supernatant was centrifuged at 8,000 rpm for 10 min and the resulting pellet was stored at -80°C until PCR analysis. A further subsample of supernatant was used for a two-step enrichment. First, 1 mL of supernatant was mixed with 9 mL of Ashdown broth [33] and incubated at 37°C. After 48 h, the Ashdown broth was mixed and 1 mL was transferred to 9 mL of erythritol broth [23] and incubated at 37°C. After 96 h, 1 mL of erythritol broth was centrifuged at 8,000 rpm for 10 min and the resulting pellet was stored at -80°C until PCR analysis.

Water samples

Water samples were collected from goat drinking troughs on farms A (n = 2) and B (n = 8). Within two hours of collection, 50 mL of each water sample were filtered sequentially through a 3 μ m filter followed by a 0.2 μ m filter. All filters were folded in four, preserved in Lysogeny Broth (LB) (10 g peptone, 5 g yeast extract, 10 g NaCl for 1 L)-glycerol 20% and stored at -80°C until analysis. For analysis, the LB-glycerol 20% was discarded and the 0.2 μ m filters were aseptically cut in half. One half was placed in 10 mL of Ashdown broth and followed the enrichment protocol described above, while the other half was stored at -80°C for PCR analysis.

DNA extraction and PCR analysis (environmental and animal samples)

Soil supernatants without pre-enrichment were extracted using the DNeasy PowerSoil Pro kit (QIAGEN, Hilden, Germany), while supernatants from enriched soil samples were extracted using the High Pure PCR Template Preparation kit (Roche, Mannheim, Germany) according to the manufacturer's instructions. The heads of the rectal swabs were cut and placed directly into the lysis buffer and extracted using the High Pure PCR Template Preparation kit. The supernatants from the enriched water samples were extracted using the same kit. The half 0.2 µm filters used to filter the water samples were ribolyzed for three cycles of 20 s at 5,500 rpm (Lysing Matrix E two mL, MPBio, Eschwege, Germany) and DNA extraction was

performed on 800 µL of the supernatant using the High Pure PCR Template Preparation kit. Fecal samples were extracted using the following protocol: 2 g of feces were weighed and placed in 20 mL sterile water, then vortexed and placed under rotary agitation for 30 minutes. After settling for 15 minutes, the supernatant was collected and subjected to double enrichment as for the soil and water samples. DNA extraction was performed as for the soil and water samples. An internal inhibition extraction control (Xeno DNA control or Diagenode IPC, Thermofisher) was added to each sample.

Initial PCR screening was performed using an *aro*A-based real-time PCR assay [34] targeting the *B. pseudomallei* complex, which includes *B. pseudomallei*, *B. thailandensis*, as well as 6 other *Burkholderia* species [35]. DNA samples that tested positive were further tested using real-time PCR assays targeting specific regions of the *B. pseudomallei* genome (*orf*11, BPSS0087, BPSS0745, and/or BPTT4176-4290) [36,37] and/or *B. thailandensis* (70 kDa) [38]. Each PCR reaction contained 5 μ L of DNA and 15 μ L of reaction mix, 1X of Universal Mastermix TaqMan Fast Advanced 2X (Applied Biosystems, Vilnius, Lithuania), 0.5 μ M of each primer, 0.1 μ M of probe, 1X of IPC Diagenode and 2.8 μ L of sterile water. The amplification procedure included an incubation at 50°C for 2 min, followed by a denaturation step at 95°C for 20 sec, then 45 cycles at 95°C for 3 sec and 60°C for 30 sec.

Culture and screening of suspect colonies

Soil enrichment broths stored at 4°C were homogenized and 100 μ L were plated onto both conventional Ashdown agar and modified *B. cepacia* CHROMagar agar (BK992 CHROMagar, Paris, France) supplemented with 4% glycerol, 500 mg/mL gentamicin and 130 mg/mL fosfomycin (CHR_GGF). Plates were incubated at 37°C and monitored daily. Suspect colonies (purple on Ashdown, green on CHR_GGF) were streaked on non-selective blood agar plates containing 5% horse serum (BA) and incubated at 37°C for 24 h to obtain a pure culture. A loopful of colonies was resuspended in sterile water, heated at 100°C for 20 min to break the cell membranes, and then centrifuged at 13,000 g for 10 min to pellet cell debris. The supernatant, rich in bacterial DNA, was used for PCR analysis (*aroA* and *orf*11 systems) to confirm colony identification. Isolates confirmed as *B. pseudomallei* by PCR were further characterized and their antimicrobial resistance patterns were identified.

B. pseudomallei strain characterization

Antibiogram. Antibiotic susceptibility testing was performed using the disk diffusion method according to the guidelines of the European Committee on Antimicrobial Susceptibility Testing (EUCAST) [39]. The antibiotics tested included: amoxicillin-clavulanate (20–10 μ g), meropenem (10 μ g), trimethoprim-sulfamethoxazole (1.25–23.75 μ g), chloramphenicol (30 μ g), imipenem (10 μ g), ceftazidime (10 μ g), and tetracycline (30 μ g) (Bio-Rad, France), as per standard testing of human samples. In addition, we tested susceptibility to colistin (10 μ g) and gentamycin (10 μ g) (Bio-Rad, France), which were used as selective antibiotics in our culture media. Fresh 0.5 McFarland suspensions were prepared and plated on Mueller-Hinton agar (Bio-Rad, France). Antibiotic discs were then placed on the agar and plates were incubated at 35°C for 18 h. EUCAST breakpoints were used for interpretation, except for colistin and gentamycin for which no EUCAST breakpoints have been published. For these two antibiotics the results are qualitative. As recommended by EUCAST, two reference strains, *Escherichia coli* ATCC 25922 and *Staphylococcus aureus* ATCC 29213, were included in the assay.

Biochemical tests. The API 20 NE system (Biomérieux, France) was used according to the manufacturer's instructions. Bacterial suspensions were adjusted to a 0.5 McFarland standard and the API cassette was then incubated at 29°C for 24 h. The web-based APIWEB

database (accessed 7 April 2023, Biomérieux, France, <u>https://apiweb.biomerieux.com/</u>) was used to interpret the profiles obtained. An oxidase test (oxidase reagent 50 x 0.75 mL, Biomérieux, France) was also performed on suspected colonies.

Multilocus sequence typing (MLST). Isolated strains were typed using MLST on 7 genes as described previously [40]. Amplified fragments were Sanger sequenced by Eurofins (Germany). Sequences were compared to the *Burkholderia pseudomallei* PubMLST database for MLST profiling (https://pubmlst.org/https://pubmlst.org/organisms/burkholderia-pseudomallei, accessed 17 February 2023) [41]. All published *B. pseudomallei* strains from South America were extracted from PubMLST: 19 were isolated from humans, 2 from animals and 2 from soil. Using BioNumerics software 7.6.3., we built a phylogenetic tree comparing the MLST profile of our strains with the 23 regional strains. The tree was built based on the 7 concatenated MLST genes and is a maximum parsimony tree with bootstrapping (n = 100).

Results

Serological survey in French Guiana and in Les Saintes

As stated in the methods, the GLANDA ELISA was originally designed for glanders diagnosis, therefore seropositivity against *B. pseudomallei* should be considered as presumptive.

In French Guiana, 16% (58/361) of cattle, 2% (2/100) of goats, 3% (4/131) of sheep, 24% (15/63) of horses and 0% (0/15) of pigs were seropositive (Table 1). The highest S/P% was observed in cattle, sheep and horses.

In Les Saintes, 39% (12/31) of goats were seropositive (<u>Table 1</u>). They were distributed over two farms (5/14 in farm A; 7/17 in farm B (<u>S1 Table</u>)), with S/P% ranging from 0 to 503%.

In French Guiana, 25 out of 56 farms tested positive, of which 10 were multi-species (Fig 3). Sixteen of the farms with positive tests were located in three communes: Macouria (n = 8), Kourou (n = 4), Sinnamary (n = 4). Interestingly, on positive farms, not all animals were positive and of the 25 GLANDA ELISA positive farms in French Guiana, 10 had animals with S/P % > 200.

Additional analysis of horses from French Guiana and follow-up of farm #24

In French Guiana, horses originating from six out of 56 farms (2 to 31 horses per farm; n = 63) were further tested by two additional methods developed for the diagnosis of glanders: the reference CFT method [16] and the newly developed Luminex method using GroEL and Hcp1 antigens [32]. All horses with positive GLANDA ELISA results were confirmed positive for Hcp1 by Luminex, but negative for CFT and GroEL by Luminex, with the exception of farm 24 where 4 horses tested positive for CFT (**Tables 2 and S1**).

Table 1.	Summar	v of serolog	zical results	(GLANDA	A ELISA) i	in domestic	animals in	French	Guiana	(n = 670) and in 1	Les Saintes	(n = 31)	1).
		,		(· */ *	,		·	- / -

Sampling site		Les Saintes				
Animal species	Cattle	Goats	Sheep	Horses	Pigs	Goats
Number of positive animals/total (%)	58/361 (16)	2/100 (2)	4/131 (3)	15/63 (24)	0/15 (0)	12/31 (39)
[Min—Max] S/P%	[0-507]	[0-99]	[0-413]	[0-435]	[0-22]	[0-503]
Number of positive farms/total (%)	17/37 (45)	2/10 (20)	2/15 (13)	4/6 (66)	0/1 (0)	2/2 (100)

S/P%: Serological titer expressed as the ratio of the sample to the positive control. Serum samples with $S/P\% \ge 70\%$ were considered positive, as per manufacturers recommendations for glanders diagnosis. Farms with at least one animal positive to the GLANDA ELISA test were considered positive.



Fig 3. Summary of GLANDA ELISA results per farm and species in French Guiana (A: cattle (n = 37), B: goats (n = 10), C: sheep (n = 15), and D: horses (n = 6)) and in Les Saintes (E: goats (n = 2)). S/P%: Serological titer expressed as the ratio of the sample to the positive control. The x-axis of the graph represents farm identification numbers ranging from 1 to 56. Each farm is assigned a unique number, and if multiple species were present on the same farm, the same number is used. The total number of animals tested per farm is shown in parentheses below each farm number. The red dotted line represents the S/P% threshold (70%). The bar above and below the box plot indicates the minimum and maximum S/P%. The top and bottom of the box indicate the 1st and 3rd quartiles of S/P%. The bars inside the box indicate the median. Due to only one pig farm being tested, we have excluded these data from the figure.

Focused study on two goat farms in Les Saintes

Serological and PCR analysis. To study the temporal evolution of the serological response in goats that initially tested ELISA-positive in November 2021, additional blood

Farm ID	Sampling date	Number of tested horses	GLANDA ELISA (P \geq 70%) Positive horses	CFT Positive horses	Luminex—Hcp1 (P \geq 43%) Positive horses	$\begin{array}{c} \text{Luminex} - \text{GroEL} \ (\text{P} \geq 45\%) \\ \text{Positive horses} \end{array}$
15	March-21	31	7	0	9	1
24	October-21	6	3	4	3	0
	January-22	10	5	4	5	0
53	March-21	2	2	0	2	0
54	March-21	11	0	0	0	0
55	March-21	4	0	0	0	0
56	April-21	5	1	0	1	0

Table 2. Serological results of 63 horses from six farms in French Guiana.

GLANDA ELISA: Enzyme-linked immunosorbent assay with the ID Screen Glanders Double Antigen Multi-species ELISA test (Innovative Diagnostics, Grabels, France). CFT: Complement fixation test. S/P%: Serological titer expressed as the ratio of the sample to the positive control. Luminex: bead-based assay targeting the recombinant proteins Hcp1 or GroEL. Thresholds used for each method are given in parentheses following the test name. Samples with anti-complementarity activity when tested by CFT (two horses on farm 15, three on farm 54 and one on farm 55) were considered as uninterpretable.

Table 3. Serological and PCR results from two goat farms in Les Saintes. The number of real-time PCR positive rectal swab samples relative to the total number tested is given for six PCR targets (one for the *B. pseudomallei* complex, four specific for *B. pseudomallei* and one specific for *B. thailandensis*). GLANDA ELISA: Enzyme-linked immunosorbent assay using the ID Screen Glanders Double Antigen Multi-species ELISA test (Innovative Diagnostics, Grabels, France). In November 2021, no rectal swabs were collected. * In March 2023, only fecal samples were collected for each goat.

Farm	Sampling date	GLANDA-ELISA (sera)		Real time PCR (rectal swabs)							
				Bpm complex			Bt				
		pos/tot (%)	[min—max] %positivity	aroA	orf11	BPSS087	BPSS0745	BPSS4208	70 kDa		
Farm A	November-21	5/14 (36)	[0-494]	not sampled							
	April-22	4/31 (13)	[0-373]	2/31	1/2	0/2	0/2	0/2	0/2		
	August-22	8/33 (24)	[0-289]	0/33	0/33	na	na	na	na		
	December-22	11/39 (28)	[0-395]	1/39	1/1	1/1	1/1	0/1	0/1		
	March-23*	7/30 (23)	[0-396]	0/30	0/30	na	na	na	na		
Farm B	November-21	November-21 7/17 (41) [0–503]		not sampled							
	April-22	7/30 (23)	[0-351]	3/30	0/3	0/3	0/3	0/3	0/3		
	August-22	5/22 (23)	[0-343]	0/22	0/22	na	na	na	na		
	December-22	7/25 (28)	[0-455]	0/25	0/25	na	na	na	na		

Bpm: Burkholderia pseudomallei, Bt: Burkholderia thailandensis. na: not analysed (samples with negative aroA and orf11 PCR analysis). pos/tot (%): number of positive animals/total (percentage), S/P%: serological titer expressed as the ratio of the sample to the positive control.

https://doi.org/10.1371/journal.pntd.0011977.t003

samples were collected in April 2022, August 2022, December 2022 on both farms A and B in Les Saintes, and in March 2023 on farm A.

Seropositive animals were consistently detected at all time points on both farms, although the percentage of positive animals varied over time. It ranged from 13% to 36% in farm A and from 23% to 41% in farm B. The highest number of ELISA-positive animals was observed in both farms in November 2021 (Table 3).

Rectal swabs were collected in parallel with blood samples to test for the fecal shedding of *B. pseudomallei*. In total, six swabs tested positive with the *aroA* PCR targeting all species of the *B. pseudomallei* complex, of which two were confirmed positive for *B. pseudomallei*: one collected in April 2022 and one in December 2022 on farm A (Table 3).

On farm A, 20 goats were sampled three times, in August and December 2022 and March 2023; 13 were consistently GLANDA ELISA negative, five were consistently GLANDA ELISA positive and two changed status during the course of the 8-month study (**Tables 4 and S1**). Of note, six goats had S/P% greater than 200% on at least two sampling occasions. One goat with the lowest S/P% (\approx 150%) later tested negative in March 2023. Conversely, two goats with initially low S/P% later had higher values. Of the seven goats that were seropositive at different times, five were over one year old. The two identified mother-daughter pairs were consistently

Table 4. Longitudinal serological study in two goat farms in Les Saintes. GLANDA ELISA: Enzyme-linked immunosorbent assay using the ID Screen Glanders Double Antigen Multi-species ELISA test (Innovative Diagnostics, Grabels, France). S/P%: Serological titer expressed as the ratio of the sample to the positive control. Negative (S/ P% < 70); Positive (S/P% $\geq 70\%$); ns: not sampled (farm B was not sampled in March 2023).

Farm	Number of animals	August 2022	November 2022	March 2023	
Farm A	5	Positive	Positive	Positive	
	1	Positive	Positive	Negative	
	1	Negative	Positive	Positive	
	13	Negative	Negative	Negative	
Farm B	1	Positive	Positive		
	2	Negative	Positive	ns	
	9	Negative	Negative		

negative. On farm B, 11 animals were sampled twice, in August and December 2022; eight were GLANDA ELISA negative twice, one was GLANDA ELISA positive twice, and two changed status during the course of the study (seroconversion). All three animals that were seropositive at least once were over one year old and had S/P% values greater than 200%.

Environmental sampling in goat pastures in Les Saintes

In December 2022, 50 soil samples and two water samples (from the drinking troughs) were collected from farm A; and 42 soil samples and eight water samples (from the drinking troughs) were collected from farm B. All samples were analyzed by PCR and bacteriological culture. The 52 samples from farm A were all PCR negative, before and after enrichment. On farm B, two (22–10884_313 and 22–10884_336) out of 42 soil samples were PCR positive for the *B. pseudomallei* complex after enrichment, but only one sample (22–10884_313) was confirmed as *B. pseudomallei* with all four PCR specific targets (Table 5). All water samples were PCR negative.

The two PCR-positive soil enrichment broths (from soils 22–10884_313 and 22– 10884_336) were plated on both Ashdown and modified CHROMagar *B. cepacia* agar media for isolation. Thirty suspect colonies from soil 22–10884_313 were PCR positive for the *B. pseudomallei* complex and all four *B. pseudomallei* specific targets, whereas the fifteen suspect colonies from soil 22–10884_336 were all PCR positive for the *B. pseudomallei* complex but negative for *B. pseudomallei* and *B. thailandensis*.

The *B. pseudomallei* colonies from soil 22–10884_313 initially appeared purple and smooth after 24 h at 37 °C on Ashdown and then changed to a rough, dry texture after 48 h (Fig 4A). Similar morphological changes were observed on modified CHROMagar agar *B. cepacia*: colonies were green and smooth after 24 h of culture, but changed to a white, rough, and dry texture after 120 h (Fig 4B).

As all of the colonies obtained from plating soil 22–10884_313 looked similar, one *B. pseu-domallei* colony (22–10884_313#20) from modified CHROMagar agar *B. cepacia* was selected for further analysis. The strain exhibited the API 20 NE profile 1-456-574 along with a positive result for the oxidase test. When compared with the profiles listed in APIWEB, the strain had the highest identification scores with *B. pseudomallei* (47.9%) and *B. cepacia* (45.3%). The antibiotic susceptibility profile of strain 22–10884_313#20 showed susceptibility to the tested antibiotics, all of which are commonly used in human melioidosis treatment (amoxicillinclavulanate, ceftazidime, tetracycline, imipenem, meropenem, trimethoprim-sulfamethoxazole, chloramphenicol, colistin, and gentamycin) (S2 Table). The MLST sequence type of strain 22–10884_313#20 was ST92. In the *Burkholderia pseudomallei* PubMLST database,

Table 5. Results of molecular testing of environmental samples from two goat farms in Les Saintes. Soil and water samples collected in December 2022 were tested by
real-time PCR for the presence of different Burkholderia spp after enrichment. Results are given as the ratio of positive samples to the total number of samples tested for
each PCR target. na: not analyzed.

Farm	Type of sample	Number of samples	Bpm complex	Врт			Bt	
			aroA	orf11	BPSS087	BPSS0745	BPSS4208	70 kDa
А	soil	50	0/50	na	na	na	na	na
	water	2	0/2	na	na	na	na	na
В	soil	42	2/42	1/2	1/2	1/2	1/2	0/2
	water	8	0/8	na	na	na	na	na

Bpm: Burkholderia pseudomallei, Bt: Burkholderia thailandensis.



Fig 4. Cultures of *Burkholderia pseudomallei* strain 22–10884_313#20. Photographs of culture plates (A) on Ashdown agar and (B) on modified *B. cepacia* CHROMagar agar (supplemented with 4% glycerol, 500 mg/mL gentamicin and 130 mg/mL fosfomycin) after a 24, 48 and 120 h incubation at 37°C. The strain 22–10884_313#20 was isolated from soil 22–10884_313 collected from farm B in Les Saintes.

ST92 is identified in nine human strains: one from a Swiss traveler to Martinique [25], one from Puerto Rico, two from Mexico [42], and five from Brazil [43]. An MLST-based tree, including all STs identified so far in South America is shown in Fig 5.

Discussion

Melioidosis has been extensively studied in established endemic regions such as Southeast Asia and Australia, with limited research in Central and South America [28,42]. In the Caribbean,



Fig 5. MLST-based phylogenetic tree of 24 *B. pseudomallei* strains from South America, including the strain 22– **10884_313#20** isolated in this study. The phylogenetic tree was constructed from 24 South American strains: 23 from the PubMLST database and the strain isolated in this study (22–10884_313#20) indicated by a red arrow, which was isolated from a goat farm in Les Saintes. The MLST tree was a maximum parsimony tree with bootstrapping (n = 100). The numbers on the branches correspond to the bootstrap values.

human cases have been documented in Guadeloupe and Martinique for over 30 years [25–29], but the local presence of *B. pseudomallei* in these areas remains unconfirmed. Isolation of strains from the environment is challenging, probably more so in regions of emergence or low endemicity. This difficulty was highlighted in a recent environmental survey in Puerto Rico, a Caribbean island, where only three out of 500 soil samples from 60 sites tested positive for *B. pseudomallei* by PCR, and only one sample was positive by culture [44].

Previous exposure to *B. pseudomallei* triggers an immune response that can be detected by serological methods [45]. While serology isn't usually favored for melioidosis epidemiology in endemic areas due to suspected frequent environmental exposure and to the challenge of interpreting positive results outside of a clinical context [46,47], serological testing shows promise in areas of emergence. In particular, for animals confined to a restricted geographical area, serological testing allows for targeted environmental investigations within the living environment of positive animals, maximizing the chance of detecting the bacterium.

In our study, we used the GLANDA ELISA, originally developed for the diagnosis of equine glanders. We extended this method to the detection of antibodies against B. pseudomallei because a previous study carried out on a goat farm in New Caledonia, where a case of melioidosis was confirmed, demonstrated the efficacy of this kit in identifying animals exposed to B. pseudomallei [45]. Moreover, its dual antigen technology allows cross-species use without the need for species-specific conjugates. Our serological screening in French Guiana and Les Saintes showed that not all farms tested had seropositive animals, regardless of the species tested. This suggests that individual serological response may vary. In French Guiana, 2% of sheep and goats, 15% of cattle and 22% of horses tested positive, while in Les Saintes, where only goats were tested, 39% tested positive. In comparison, reported prevalences in endemic countries using other serological methods vary, such as 6% and 13.6% in sheep, 0.3% and 2.6% in goats, and 2% and 7.6% in cattle in Thailand [47] and Malaysia [46], respectively. However, differences in melioidosis endemicity and serological testing methods between countries make direct comparisons difficult. Also, our study included only two goat herds in Les Saintes and used opportunistic sampling in French Guiana, a sampling method that is not ideal to estimate prevalences.

In order to confirm our serology results, we carried out surveillance on two goat farms with seropositive animals in Les Saintes. Molecular screening in conjunction with an environmental survey confirmed fecal shedding of *B. pseudomallei* on one farm and successfully led to the isolation of an environmental strain of *B. pseudomallei* on the other farm. This confirms the usefulness of the GLANDA ELISA kit in animals to assess possible past environmental exposure to *B. pseudomallei*. However, due to the possibility of antibody cross-reactivity [48,49], direct detection of the bacterium is needed for conclusive evidence.

There is limited data on *B. pseudomallei* concentrations in the environment, but it's likely that the bacteria are present at low concentrations in areas of emergence or low endemicity, making detection more difficult. To increase the chance of detection, we optimized the protocol from existing guidelines [22] by testing two new media in addition to the conventionally recommended Ashdown medium [33]. First, an erythritol-based medium was used as the sole carbon source in the second step of a two-step enrichment to select for *B. pseudomallei* [23]. A similar two-step enrichment protocol (TBSS-C50 + erythritol) was recently used to isolate environmental *B. pseudomallei* strains in Ghana, West Africa [24]. Second, we used a chromogenic medium CHROMagar *B. cepacia* originally developed for the detection of *B. cepacia* in clinical samples [50]. To increase its selectivity for *B. pseudomallei*, the medium was modified by incorporating new antibiotics. We confirmed that reference strains of *B. pseudomallei* grew well on this modified medium (S1 Fig). To our knowledge, this is the first use of a chromogenic medium to improve the selection of *B. pseudomallei* colonies from poly-contaminated

environmental matrices. Despite these protocol improvements, in our study only one soil sample (out of 92 samples collected from seropositive farms) was found positive for *B. pseudomallei* by PCR, with a single strain isolated. Although the use of CHROMagar *B. cepacia* media appears promising, further protocol optimization (e.g., sample volume, enrichment time, chromogenic and/or selective medium for environmental matrices) will be required to facilitate the detection and isolation of environmental strains of *B. pseudomallei*. Of note, an additional soil sample was PCR positive for the *B. pseudomallei* complex but not for *B. pseudomallei* or *B. thailandensis* species. Further characterization of all non-*pseudomallei Burkholderia spp.* strains isolated in this study will help identify all the environmental *Burkholderia* species to which animals may be exposed and to assess their cross-reactivity with the GLANDA ELISA test.

Our study provides the first definitive confirmation of the local establishment of *B. pseudo-mallei* in Les Saintes, where two cases of human melioidosis were reported in 1997 (a tourist visiting Les Saintes for 3 weeks) [29] and in 2017 (a local resident) [27]. The MLST profile of the strain isolated in our study (ST92) was identical to that of the 1997 human case [29]. In the *Burkholderia pseudomallei* PubMLST database, ST92 was identified in nine human strains: one from a Swiss traveler to Martinique [25], one from Puerto Rico, two from Mexico [8] and five from Brazil [7], illustrating the spatial clustering of this sequence type.

Our study is subject to several limitations related to the use of the GLANDA ELISA test. A major uncertainty was the identity of the recombinant protein used in this commercial kit that has been validated for the diagnosis of equine glanders. Although we speculate that this undisclosed protein is also present in *B. pseudomallei* (due to the very close relationship between *B. mallei* and *B. pseudomallei*), it is possible that the antibodies detected by the test could be triggered by other environmental *Burkholderia* species, particularly those belonging to the *B. pseudomallei* complex. The presence of *B. thailandensis* in Les Saintes and French Guiana is uncertain, although it has recently been detected in the United States [48]. Therefore, the results obtained with this GLANDA ELISA kit are only an indication of the possible exposure of the animals to *B. pseudomallei*. In areas of emergence or low endemicity with little to no clinical cases, only the isolation of the bacterium can confirm the serological results.

Another limitation is related to equids and glanders. The distinction between glanders (caused by *B. mallei*) and melioidosis (caused by *B. pseudomallei*) is difficult due to their high antigenic similarity [17], with *B. mallei* being a monophyletic clade within *B. pseudomallei* [49]. As glanders is frequently reported in Brazil [51], a country bordering French Guiana, the detection of GLANDA ELISA positive horses in French Guiana warranted further investigation. Clinically healthy horses that tested GLANDA ELISA positive were all Hcp1 positive by Luminex but GroEL negative by Luminex, both proteins being key markers for the diagnosis of glanders [32]. When CFT (the reference method for the diagnosis of glanders) was added as a fourth test, all GLANDA ELISA positive horses were CFT negative, with the exception of horses on farm #24. On this farm, the detection of horses positive to three of the four tests relevant to the diagnosis of glanders led to a follow-up examination three months later, which confirmed the initial serological results and the absence of clinical signs typical of acute glanders (e.g., respiratory distress or abscesses). Despite the absence of previously reported cases of glanders or melioidosis in French Guiana, the ultimate health status of these equines remains uncertain given the chronic potential of these two diseases [35,52,53].

In conclusion, our study supports the hypothesis that animals could be used as sentinels to detect the presence of *B. pseudomallei* in the environment, with the GLANDA ELISA test being a relevant tool for this purpose in areas of emergence or where the endemicity of melioidosis is uncertain. However, definitive proof must be obtained by direct detection of the bacteria. Further research is also needed to establish the specificity of this test in contexts other than

those in which it was originally designed and validated. In addition, the threshold used in our study was originally established for equine glanders in a clinical setting and may require reevaluation to assess environmental exposure to *B. pseudomallei* in different host species. Such refinement may be particularly useful in areas of emergence or low endemicity. This study confirmed the presence of the bacterium in Les Saintes. Similar serological and environmental studies now need to be carried out in Guadeloupe and Martinique, where human cases have also been reported [25–30]. Although no cases of melioidosis have been reported in French Guiana, our serological results suggest the presence of *B. pseudomallei* and the need for environmental surveys around the ELISA-positive farms identified in this study and for retrospective re(evaluation) of human infections of unconfirmed ethiology and with symptoms similar to those of melioidosis.

Supporting information

S1 Table. Serological results of animals in French Guiana and Les Saintes (Guadeloupe). (XLSX)

S2 Table. Antibiogram results of 22–10884_313#20 strain. (XLSX)

S1 Fig. Evaluation of CHROMagar *B. cepacia* as a relevant media for other *Burkholderia* strains.

(TIF)

Acknowledgments

We thank all the livestock farmers, especially the owners of the two goat farms in Les Saintes, who allowed repeated sampling to complete the longitudinal study. Special thanks to CHRO-Magar for supplying the culture media and to Dr. Sophie Granier for her valuable advice on the implementation of the antibiogram test. We thank Carina Hall and Nathan Stone for help-ful early discussions regarding environmental samplings.

Author Contributions

Conceptualization: Vanina Guernier-Cambert, Sébastien Breurec, Emma Rochelle-Newall, Karine Laroucau.

Investigation: Mégane Gasqué, Gil Manuel, Rachid Aaziz, Jules Terret, Thomas Deshayes.

Methodology: Mégane Gasqué, Karine Laroucau.

Resources: Gil Manuel, Xavier Baudrimont, Sébastien Breurec.

Supervision: Vanina Guernier-Cambert, Emma Rochelle-Newall, Karine Laroucau.

Writing – original draft: Mégane Gasqué, Karine Laroucau.

Writing – review & editing: Mégane Gasqué, Vanina Guernier-Cambert, Gil Manuel, Rachid Aaziz, Jules Terret, Thomas Deshayes, Xavier Baudrimont, Sébastien Breurec, Emma Rochelle-Newall, Karine Laroucau.

References

 Meumann EM, Limmathurotsakul D, Dunachie SJ, Wiersinga WJ, Currie BJ. Burkholderia pseudomallei and melioidosis. Nat Rev Microbiol. 2023. https://doi.org/10.1038/s41579-023-00972-5 PMID: 37794173

- Limmathurotsakul D, Golding N, Dance DAB, Messina JP, Pigott DM, Moyes CL, et al. Predicted global distribution of *Burkholderia pseudomallei* and burden of melioidosis. Nat Microbiol. 2016; 1: 1–5. <u>https:// doi.org/10.1038/nmicrobiol.2015.8 PMID: 26877885</u>
- Steinmetz I, Wagner G, Kanyala E, Sawadogo M, Soumeya H, Teferi M, et al. Melioidosis in Africa: Time to Uncover the True Disease Load. Trop Med Infect Dis. 2018; 3. <u>https://doi.org/10.3390/</u> tropicalmed3020062 PMID: 30274458
- Birnie E, Wiersinga WJ, Limmathurotsakul D, Grobusch MP. Melioidosis in Africa: should we be looking more closely? Future Microbiol. 2015; 10: 273–281. https://doi.org/10.2217/fmb.14.113 PMID: 25689538
- Petras JK, Elrod MG, Ty MC, Dawson P, O'Laughlin K, Gee JE, et al. Locally Acquired Melioidosis Linked to Environment—Mississippi, 2020–2023. N Engl J Med. 2023; 389: 2355–2362. https://doi.org/ 10.1056/NEJMoa2306448 PMID: 38118023
- Engelthaler DM, Bowers J, Schupp JA, Pearson T, Ginther J, Hornstra HM, et al. Molecular Investigations of a Locally Acquired Case of Melioidosis in Southern AZ, USA. PLoS Negl Trop Dis. 2011; 5: e1347. https://doi.org/10.1371/journal.pntd.0001347 PMID: 22028940
- Rolim DB, Lima RXR, Ribeiro AKC, Colares RM, Lima LDQ, Rodríguez-Morales AJ, et al. Melioidosis in South America. Trop Med Infect Dis. 2018; 3: 60. <u>https://doi.org/10.3390/tropicalmed3020060</u> PMID: 30274456
- Sanchez-Villamil JI, Torres AG. Melioidosis in Mexico, Central America, and the Caribbean. Trop Med Infect Dis. 2018; 3: 24. https://doi.org/10.3390/tropicalmed3010024 PMID: 29780897
- Rolim DB, Feitosa Lima Vilar DC, Queiroz Sousa A, Miralles IS, De Oliveira DCA, Harnett G, et al. Melioidosis, Northeastern Brazil. Emerg Infect Dis. 2005; 11: 3. https://doi.org/10.3201/eid1109.050493 PMID: 16229782
- Gassiep I, Armstrong M, Norton R. Human melioidosis. Clin Microbiol Rev. 2020; 33: e00006–19. https://doi.org/10.1128/CMR.00006-19 PMID: 32161067
- Sprague LD, Neubauer H. Melioidosis in Animals: A review on epizootiology, diagnosis and clinical presentation. J Vet Med Ser B. 2004; 51: 305–320. https://doi.org/10.1111/j.1439-0450.2004.00797.x PMID: 15525357
- Hambali I, Abba Y, Bitrus A, Peter I, Jesse F, Balakrishnan T, et al. Seroprevalence of melioidosis in sheep and goats from selected small ruminants in Selangor, Malaysia. Adv Anim Vet Sci. 2018; 2: 88– 94. https://doi.org/10.17582/journal.aavs/2018/6.2.88.94
- Musa HI, Hassan L, Rachmat R, Chandrawthani P, Zunita Z, Saleha A. Seroprevalence of melioidosis among livestock in Malaysia from 2000–2009. Malays J Vet Res. 2012; 3: 41–46.
- Duval BD, Elrod MG, Gee JE, Chantratita N, Tandhavanant S, Limmathurotsakul D, et al. Evaluation of a latex agglutination assay for the identification of *Burkholderia pseudomallei* and *Burkholderia mallei*. Am J Trop Med Hyg. 2014; 90: 1043–1046. https://doi.org/10.4269/ajtmh.14-0025 PMID: 24710616
- Elschner MC, Melzer F, Singha H, Muhammad S, Gardner I, Neubauer H. Validation of a commercial glanders ELISA as an alternative to the CFT in international trade of equidae. Front Vet Sci. 2021; 8: 628389. https://doi.org/10.3389/fvets.2021.628389 PMID: 33665218
- Elschner MC, Laroucau K, Singha H, Tripathi BN, Saqib M, Gardner I, et al. Evaluation of the comparative accuracy of the complement fixation test, Western blot and five enzyme-linked immunosorbent assays for serodiagnosis of glanders. Plos One. 2019; 14: e0214963. <u>https://doi.org/10.1371/journal.pone.0214963</u> PMID: 30951554
- Wernery U, Caveney M, Wernery R, Raghavan R, Laroucau K, Syriac G, et al. Evaluation of serological responses in horses challenged with *Burkholderia pseudomallei* using current diagnostic tests for glanders. Vet Ital. 2019; 55: 261–267. https://doi.org/10.12834/VetIt.1701.9026.2 PMID: 31599551
- Kaestli M, Mayo M, Harrington G, Ward L, Watt F, Hill J, et al. Landscape Changes Influence the Occurrence of the Melioidosis Bacterium *Burkholderia pseudomallei* in Soil in Northern Australia. PLoS Negl Trop Dis. 2009; 3. https://doi.org/10.1371/journal.pntd.0000364 PMID: 19156200
- Zimmermann RE, Ribolzi O, Pierret A, Rattanavong S, Robinson MT, Newton PN, et al. Rivers as carriers and potential sentinels for *Burkholderia pseudomallei* in Laos. Sci Rep. 2018; 8: 8674. <u>https://doi.org/10.1038/s41598-018-26684-y PMID: 29875361</u>
- Canales R, Sanchez-Okrucky R, Bustamante L, Vences M, Dennis M. Melioidosis in a bottlenose dolphin (*Tursiops truncatus*) after a hurricane in the Caribbean islands. J Zoo Wildl Med. 2020; 51: 443– 447. https://doi.org/10.1638/2019-0050 PMID: 32549577
- Guendel I, Ekpo LL, Hinkle MK, Harrison CJ, Blaney DD, Gee JE, et al. Melioidosis after Hurricanes Irma and Maria, St. Thomas/St. John District, US Virgin Islands, October 2017. Emerg Infect Dis. 2019; 25: 1952–1955. https://doi.org/10.3201/eid2510.180959 PMID: 31538918

- Limmathurotsakul D, Dance DAB, Wuthiekanun V, Kaestli M, Mayo M, Warner J, et al. Systematic review and consensus guidelines for environmental sampling of *Burkholderia pseudomallei*. PLoS Negl Trop Dis. 2013; 7: e2105. https://doi.org/10.1371/journal.pntd.0002105 PMID: 23556010
- Trinh TT, Assig K, Tran QTL, Goehler A, Bui LNH, Wiede C, et al. Erythritol as a single carbon source improves cultural isolation of *Burkholderia pseudomallei* from rice paddy soils. PLoS Negl Trop Dis. 2019; 13. https://doi.org/10.1371/journal.pntd.0007821 PMID: 31634353
- Oduro G, Robberts FJL, Dartey PKA, Owusu-Ofori A, Oppong C, Gyampomah TK, et al. On the Environmental Presence of *Burkholderia pseudomallei* in South-Central Ghana. Appl Environ Microbiol. 2022; 88: e00600–22. https://doi.org/10.1128/aem.00600-22 PMID: 35652663
- Gétaz L, Michel Y, Merlani P, Emonet S, Abbas M, Loutan L, et al. Fatal acute melioidosis in a tourist returning from Martinique Island, November 2010. Euro Surveill. 2010; 2011: pii = 19758.
- Meckenstock R, Therby A, Marque-Juillet S, Monnier S, David Khau D, Pangon B, et al. Cutaneous melioidosis in adolescent returning from Guadeloupe. Emerg Infect Dis. 2012; 18: 359–360. <u>https://doi.org/10.3201/eid1802.111603</u> PMID: 22305492
- Melot B, Bastian S, Dournon N, Valade E, Gorgé O, Le Fleche A, et al. Three new cases of melioidosis, Guadeloupe, French West Indies. Emerg Infect Dis. 2020; 26: 617–619. <u>https://doi.org/10.3201/</u> eid2603.190718 PMID: 32091384
- 28. Pasquier J, Olive C, Hurtrel G, Turmel J-M, Cabié A. Melioidosis. Lancet Infect Dis. 2019; 19: 1057. https://doi.org/10.1016/S1473-3099(19)30481-5 PMID: 31559959
- Pérez JM, Petiot A, Adjidé C, Gerry F, Goursaud R, Juminer B. First case report of melioidosis in Guadeloupe, a French West Indies archipelago. Clin Infect Dis Off Publ Infect Dis Soc Am. 1997; 25: 164–165. https://doi.org/10.1086/516896 PMID: 9243060
- Olive C, Loetitia G, Desbois N, Roche B, Jouannelle J, Dodin A. Septic pyemic form of human melioidosis: a first case in the French Antilles. Presse Medicale Paris Fr 1983. 1995; 24: 1270.
- Laroucau K, Colaneri C, Jaÿ M, Corde Y, Drapeau A, Durand B, et al. Interlaboratory ring trial to evaluate CFT proficiency of European laboratories for diagnosis of glanders in equids. Vet Rec. 2016; 178: 632–632. https://doi.org/10.1136/vr.103617 PMID: 27122499
- Laroucau K, Saqib M, Martin B, Deshayes T, Bertin C, Wernery U, et al. Development of a microsphere-based immunoassay for the serological detection of glanders in equids. Acta Trop. 2020; 207: 105463. https://doi.org/10.1016/j.actatropica.2020.105463 PMID: 32302692
- Ashdown LR, Clarke SG. Evaluation of culture techniques for isolation of *Pseudomonas pseudomallei* from soil. Appl Environ Microbiol. 1992; 58: 4011–4015.
- Laroucau K, Aaziz R, Vorimore F, Varghese K, Deshayes T, Bertin C, et al. A genetic variant of *B. mallei* detected in Kuwait: consequences for the PCR diagnosis of glanders. Transbound Emerg Dis. 2020; 1– 4. https://doi.org/10.1111/tbed.13777 PMID: 33245616
- Hall CM, Baker AL, Sahl JW, Mayo M, Scholz HC, Kaestli M, et al. Expanding the Burkholderia pseudomallei complex with the addition of two novel species: Burkholderia mayonis sp. nov. and Burkholderia savannae sp. nov. Appl Environ Microbiol. 2022; 88: e01583–21. https://doi.org/10.1128/AEM.01583-21 PMID: 34644162
- 36. Göhler A, Trung TT, Hopf V, Kohler C, Hartleib J, Wuthiekanun V, et al. Multitarget quantitative PCR improves detection and predicts cultivability of the pathogen *Burkholderia pseudomallei*. Appl Environ Microbiol. 2017; 83: e03212–16. https://doi.org/10.1128/AEM.03212-16 PMID: 28188208
- Thibault FM, Valade E, Vidal DR. Identification and discrimination of *Burkholderia pseudomallei*, *B. mallei*, and *B. thailandensis* by real-time PCR targeting type III secretion system genes. J Clin Microbiol. 2004; 42: 5871–5874. https://doi.org/10.1128/JCM.42.12.5871–5874.2004
- Lowe C-W, Satterfield BA, Nelson DB, Thiriot JD, Heder MJ, March JK, et al. A quadruplex real-time PCR assay for the rapid detection and differentiation of the most relevant members of the *B. pseudo-mallei complex*. *B. mallei, B. pseudomallei,* and *B. thailandensis*. Plos One. 2016; 11: e0164006. https://doi.org/10.1371/journal.pone.0164006 PMID: 27736903
- Matuschek E, Brown DFJ, Kahlmeter G. Development of the EUCAST disk diffusion antimicrobial susceptibility testing method and its implementation in routine microbiology laboratories. Clin Microbiol Infect. 2014; 20: O255–O266. https://doi.org/10.1111/1469-0691.12373 PMID: 24131428
- Godoy D, Randle G, Simpson A, Aanensen D, Pitt T, Kinoshita R, et al. Multilocus sequence typing and evolutionary relationships among the causative agents of melioidosis and glanders, *Burkholderia pseudomallei* and *Burkholderia mallei*. J Clin Microbiol. 2003; 41: 2068–2079. <u>https://doi.org/10.1128/JCM</u>. 41.5.2068–2079.2003
- **41.** Jolley KA, Bray JE, Maiden MCJ. Open-access bacterial population genomics: BIGSdb software, the PubMLST.org website and their applications. Wellcome Open Res. 2018; 3: 124. https://doi.org/10. 12688/wellcomeopenres.14826.1 PMID: 30345391

- **42.** Cheng JW, Hayden MK, Singh K, Heimler I, Gee JE, Proia L, et al. *Burkholderia pseudomallei* infection in US traveler returning from Mexico, 2014. Emerg Infect Dis. 2015; 21: 1884–1885. <u>https://doi.org/10.3201/eid2110.150815 PMID: 26401597</u>
- Gee JE, Gulvik CA, Castelo-Branco DSCM, Sidrim JJC, Rocha MFG, Cordeiro RA, et al. Genomic diversity of *Burkholderia pseudomallei* in Ceara, Brazil. mSphere. 2021; 6: e01259–20. <u>https://doi.org/</u> 10.1128/mSphere.01259-20 PMID: 33536328
- 44. Hall CM, Jaramillo S, Jimenez R, Stone NE, Centner H, Busch JD, et al. Burkholderia pseudomallei, the causative agent of melioidosis, is rare but ecologically established and widely dispersed in the environment in Puerto Rico. PLoS Negl Trop Dis. 2019; 13: e0007727. <u>https://doi.org/10.1371/journal.pntd.</u>0007727 PMID: 31487287
- 45. Desoutter A, Deshayes T, Vorimore F, Klotoe B, Durand B, Colot J, et al. Isolation of *Burkholderia pseudomallei* from a goat in New Caledonia: implications for animal and human health monitoring and sero-logical tool comparison. BMC Vet Res. 2024; 20: 114. <u>https://doi.org/10.1186/s12917-024-03957-5</u> PMID: 38521903
- 46. Masrin A, Fatiha A, Fakhrulisham R, Sharil Azwan M. Geographical mapping on seropositive status of melioidosis among livestock in Malaysia from 2012 to 2016. Malays J Vet Res. 2018; 9: 44–52.
- 47. Srikawkheaw N, Lawhavinit O. Detection of antibodies against melioidosis from animal sera in Thailand by indirect haemagglutination test. Kasetsart J—Nat Sci. 2007; 41: 81–85.
- Hall CM, Stone NE, Martz M, Hutton SM, Santana-Propper E, Versluis L, et al. Burkholderia thailandensis Isolated from the Environment, United States. Emerg Infect Dis. 2023; 29: 618–621. https://doi.org/ 10.3201/eid2903.221245 PMID: 36823515
- Losada L, Ronning CM, DeShazer D, Woods D, Fedorova N, Stanley Kim H, et al. Continuing evolution of *Burkholderia mallei* through genome reduction and large-scale rearrangements. Genome Biol Evol. 2010; 2: 102–116. https://doi.org/10.1093/gbe/evq003 PMID: 20333227
- **50.** Maruri-Aransolo A, de Dios Caballero J, Michelena M, Medina-Pascual MJ, Carrasco G, Asensio O, et al. Evaluation of CHROMagar[™] *B. cepacia* agar for the detection of *Burkholderia cepacia* complex species from sputum samples of patients with cystic fibrosis. Eur J Clin Microbiol Infect Dis. 2024 [cited 27 Jun 2024]. https://doi.org/10.1007/s10096-024-04845-4 PMID: 38780755
- Mota RA, Junior JWP. Current status of glanders in Brazil: recent advances and challenges. Braz J Microbiol. 2022; 53: 2273–2285. https://doi.org/10.1007/s42770-022-00814-1 PMID: 36044120
- Bourrier M. Mélioïdose Equine en Mayenne dans une écurie de chevaux de selle. Bulletin de la Société vétérinaire pratique de France. Bulletin de la Société vétérinaire pratique de France. 1978. pp. 673– 679.
- 53. Singha H, Shanmugasundaram K, Tripathi BN, Saini S, Khurana SK, Kanani A, et al. Serological surveillance and clinical investigation of glanders among indigenous equines in India from 2015 to 2018. Transbound Emerg Dis. 2020; 67: 1336–1348. https://doi.org/10.1111/tbed.13475 PMID: 31916415